Author's response to reviews

Title: The radiosensitizing effects of doranidazole on human colorectal cancer cells exposed to high dose irradiation

Authors:

Li Zhang (tyouri19652004@yahoo.com.cn)
Aimin Gong (yinke760109@163.com)
Jun Ji (cancer-1976@hotmail.com)
Yuanyuan Wu (wuyuanyuan9@yahoo.com.cn)
Xiaoyu Zhu (zhuxiaoyu66@yahoo.com.cn)
Suqing Lv (lusuqing341@hotmail.com)
Honazhu Lv (pole163@163.com)
Xizhuo Sun (xizhuomd@163.com)

Version: 7 Date: 4 October 2007

Author's response to reviews: see over
Dear Editor

Thank you very much for accepting our paper to publish on the BMC Cancer, and we have made the changes as you indicated.

We want to know the publish form of our paper (online style only or print style)?

Sincerely yours,

4, Oct 2007

Li Zhang M.D., Ph.D
Department of Central Laboratory, Dalian Municipal Central Hospital, Xinan Road 826, Shahekou-qu, Dalian, 116033, China
Phone: + 86-411-84441341, FAX: + 86-411-84441341

E-mail address: tyouri19652004@yahoo.com.cn